Author’s response to reviews

Title: The prevalence and types of discordance between physician perception and objective data from standardized measures of RA disease activity in real-world clinical practice in the US

Authors:

Wenhui Wei (wenhui.wei@regeneron.com)
Emma Sullivan (emma.sullivan@adelphigroup.com)
Stuart Blackburn (stuart.blackburn@adelphigroup.com)
Chieh-I Chen (chieh-i.chen@regeneron.com)
James Piercy (james.piercy@adelphigroup.com)
Jeffrey Curtis (jrcurtis@uabmc.edu)

Version: 4 Date: 01 Apr 2019

Author’s response to reviews:

Dear Ms. Wicker

Thank you for your recent email advising of some needed corrections on our manuscript.

Please find in the submitted draft that we have addressed all the outstanding issues:

1. Detailed ethics statement in the Ethical approval and consent to participate section.

2. Clarification in the Funding Section of the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

3. No marked manuscript submitted, but only a clean version.
I hope the new version is acceptable for publication.

Thank you again

Sincerely,

Wenhui Wei (on behalf of all authors)